

AI translation • View original & related papers at chinarxiv.org/items/chinaxiv-202508.00031

Review of the development and application of high flux reactors

Authors: Li, Jian, Xu, Wei, She, Ding, Xie, Heng, Liu, Zhihong, Shi, Lei, She,

Ding

Date: 2025-07-29T09:55:04+00:00

Abstract

High flux reactors (HFRs) are a special type of research reactor aimed at providing a high neutron flux. Compared with power reactors and other research reactors, HFRs have unique technical features in terms of reactor core design, irradiation capability, and operating characteristics. They can be applied to the irradiation tests of nuclear fuels and materials, radioisotope production, neutron science, and experiments. This paper reviews HFRs, including their development history, technical features, and application areas, as well as trends in the development of new and advanced HFRs.

Full Text

Preamble

Review of the Development and Application of High Flux Reactors

Jian Li¹, Wei Xu², Ding She¹,*, Heng Xie¹, Zhi-Hong Liu¹, Lei Shi¹

¹Institute of Nuclear and New Energy Technology, Collaborative Innovation Center of Advanced Nuclear Energy Technology, Key Laboratory of Advanced Reactor Engineering and Safety of Ministry of Education, Tsinghua University, Beijing 100084, China

²School of Nuclear Science and Engineering, Beijing Key Laboratory of Passive Safety Technology for Nuclear Energy, North China Electric Power University, Beijing, 102206, China

*Corresponding author

E-mail address: sheding@tsinghua.edu.cn

Abstract: High flux reactors (HFRs) are a special type of research reactor designed to provide high neutron flux. Compared with power reactors and other research reactors, HFRs possess unique technical features in reactor core design,



irradiation capability, and operating characteristics. They can be applied to irradiation tests of nuclear fuels and materials, radioisotope production, neutron science, and various experiments. This paper reviews HFRs, including their development history, technical features, application areas, and trends in the development of new and advanced HFRs.

Keywords: high flux reactor, development, design features, application fields, review

1 Introduction

Unlike power reactors that generate electricity, heat, or propulsion, research reactors [1,2] are primarily used for scientific research, engineering tests, or other applications utilizing neutrons, gamma rays, or other radiations [3-5]. Neutron flux represents one of the most important parameters of a research reactor. Research reactors with neutron fluxes of 10^1 - 10^1 n/(cm²·s) or higher are commonly categorized as high flux reactors (HFRs) [6]. Notably, although pulsing reactors can provide these neutron flux values on certain timescales—for instance, IBR-2 can deliver an average fast neutron flux of 2.26×10^1 n/(cm²·s) [7]—they fall outside the scope of this discussion, which focuses on research reactors providing stable neutron fluxes.

Compared with other research reactors, HFRs incorporate specific technical features to achieve substantially higher intensities of fast or thermal neutrons, which play a crucial role in investigations of new fuel elements and materials science, theoretical and applied physics research, and production of rare isotopes. Testing materials for nuclear energy applications, conducting fundamental physics research, and producing industrial and medical isotopes all require high-level neutron fluxes. Driven by these requirements, HFRs with one or more of these functions have been designed and constructed in various countries.

According to statistics from the International Atomic Energy Agency (IAEA) website [8], 840 research reactors are distributed across 70 countries and regions worldwide. As shown in Table 1 , most existing research reactors are being decommissioned or have already been decommissioned, with only 225 research reactors currently operating. Presently, the country with the largest number of operational research reactors (including criticality experimental devices) is Russia (54 reactors), followed by the USA (49 reactors), China (17 reactors), India (5 reactors), Germany (5 reactors), and Canada (4 reactors). Research reactors used worldwide are summarized in Fig. 1 [Figure 1: see original paper]. Among them, 40 reactors (representing 17.8%) have neutron fluxes higher than $10^1 \text{ n/(cm}^2 \cdot \text{s})$, which fall within the scope of HFRs.

Most operational HFRs were built 50 years ago and face problems such as aging and decommissioning. Therefore, new and advanced HFRs must be developed in the near future. This paper reviews HFRs, covering their historical development, technical features, and application areas, while attempting to identify future trends for developing next-generation HFRs.



The remainder of this paper is organized as follows: Section 2 reviews the development history of HFRs. Section 3 analyzes and summarizes the technical features of HFRs, including reactor core design, irradiation capability, and operating characteristics. Section 4 discusses the main applications of HFRs. Finally, Section 5 discusses future trends for HFRs.

2 Historical Development of High Flux Reactors

After the first nuclear reactor, Chicago Pile 1 (CP-1), achieved criticality in 1942, humanity entered a new era of nuclear energy. In 1943, a CP-1 reactor called CP-2 was reconstructed, fueled with natural uranium and moderated by graphite. The normal maximum power of CP-2 was 1 kW, though it could also operate at 5 or 10 kW for short periods in certain experiments [1]. CP-2 is considered the first research reactor, and early experiments performed in this reactor supported the development of subsequent reactors. Early research reactors included X-10, CP-3, LOPO, and HYPO [2]. These reactors were used to conduct extensive experimental programs related to reactor prototype design and operating characteristics of this reactor type. The operating power of these early research reactors typically ranged from kW to MW, with relatively low neutron flux.

With the development of nuclear energy, the demand for higher neutron fluxes for material testing and other applications gradually increased. In 1952, the Material Testing Reactor (MTR) [9] in the United States became critical. This reactor had a power of 40 MW with an average thermal neutron flux of 3×10^1 n/(cm²·s). Based on the neutron flux level achieved by the MTR, this was the first true HFR. In the following decades, multiple HFRs were designed and constructed, including SM [10], High Flux Isotope Reactor (HFIR) [11], Advanced Test Reactor (ATR) [12], High Flux Engineering Test Reactor (HFETR) [13], China Advanced Research Reactor (CARR) [14], and Open Pool Australian Light-water reactor (OPAL) [15]. Based on technical development and advancement levels, the historical development of HFRs can be roughly divided into three stages: the early stage (1940s–1960s), mid-term stage (1960s–2000s), and current stage (from the 2000s to present), as shown in Fig. 2 [Figure 2: see original paper].

Representative HFRs from different stages are summarized in Table 2 [2,8,16-24], which includes their power level, reactor type, status, fuel assembly, neutron flux, first criticality time, and estimated decommissioning time.

2.1 Early Stage

The early stage of HFR development spanned from the 1940s to the 1960s, with typical representatives including the MTR and SM-2. During this period, HFRs underwent significant technological improvements based on typical research reactors to achieve higher neutron fluxes and broader applications, such as the MTR-type fuel assembly and methods for obtaining ultra-high neutron flux in



the reactor core.

The design of MTR began in 1944 at Oak Ridge National Laboratory (ORNL), USA, primarily to satisfy requirements for higher neutron flux in material irradiation tests. MTR achieved its first criticality on March 31, 1952. The reactor' s thermal power was 30 MW, and it adopted highly enriched uranium (HEU) fuel. The reactor was cooled and moderated with light water, and beryllium and graphite were used as reflectors. The operating power was increased to 40 MW in 1955, resulting in an average thermal neutron flux of 3×10^1 n/(cm²·s), which maintained the highest neutron flux for many years. Plate fuel assemblies were utilized to permit higher power density and sufficient cooling. This type of fuel consists of an aluminum-uranium mixture with aluminum cladding and is called MTR-type reactor fuel [25]. The MTR-type fuel had a profound impact on the development of research reactor fuel assemblies, and many fuel assemblies used in later HFRs, such as CARR and JRR-3M, are based on this design scheme. A cross-sectional view of the MTR-type fuel assembly is depicted in Fig. 3 [Figure 3: see original paper]. The irradiation application design of MTR was also flexible, with over 100 irradiation positions and six horizontal neutron beam tubes.

The SM-2 reactor [2,22] was the first epithermal neutron spectrum research reactor with water as a moderator. It was a vessel-type HFR in the Soviet Union with a power of 100 MW. SM-2 had 15 vertical channels for radiation resistance testing, four vertical high-temperature channels for corrosion resistance testing, six vertical channels for the accumulation of transuranic elements and radioisotopes, and five horizontal channels. Construction of SM-2 began in January 1956, and it achieved first criticality on January 10, 1961. The power density of SM-2 was extremely high (the estimated average value was 2×10^3 MW/m³); therefore, an ultrahigh neutron flux was achieved in the in-core irradiation channels. At that time, the thermal neutron flux in SM-2 was 3.3×10^1 n/(cm²·s) at a thermal power of 50 MW. The construction and successful operation of SM-2 motivated the construction of HFRs in the United States for hard neutron spectrum applications, including the High Flux Beam Reactor (HFBR) and HFIR. The SM-2 reactor was characterized by the following features in its early design: high volumetric inhomogeneity coefficients (~6), deep reactivity losses due to Xe-135 poisoning (>4%), and significant burnup reactivity temperature losses. In 1965, the fuel element of SM-2 was changed from plate fuel to cruciform fuel. In 1974, the thermal neutron flux in the neutron flux trap was improved to 5.0×10^1 n/(cm²·s) when the nominal thermal power was increased to 100 MW. From 1984 to 1987, the horizontal channel of the reactor was "closed." SM-2 was finally inherited by Russia and updated to SM-3 in 1991-1992, as discussed later.

2.2 Mid-term Stage

The mid-term development stage of HFRs refers to the period from the 1960s to the 2000s. During this period, HFR technology matured, and several high



flux research reactors were successively built, including the ATR, HFIR, and HFETR.

Although MTR consistently provided a high neutron flux, the pursuit of even higher neutron flux continued. Based on this background, an improved test reactor design effort began in 1955. The power level was increased to achieve higher neutron flux in irradiation test positions, and experimental loops were installed for fuel irradiation. The new test reactor was called ETR, which was later renamed ATR [12]. The critical device of ATR (ATRC) achieved its first criticality in 1964, and ATR was put into operation in 1967. ATR has a maximum power of 250 MW, with maximum thermal and fast neutron fluxes of $1.0 \times 10^1 \text{ n/(cm}^2 \cdot \text{s)}$ (E < 0.625 eV) and $5.0 \times 10^1 \text{ n/(cm}^2 \cdot \text{s)}$ (E > 1 MeV), respectively. As shown in Fig. 4 [Figure 4: see original paper], arc-type fuel assemblies fabricated from UAlx fuel meat with HEU (90 wt%) were adopted, and the reactor core comprises 40 fuel assemblies. In contrast to the widely used control rods in other HFRs, ATR adopted rotation control drums, which helped form a relatively uniform axial power distribution. Nine neutron flux traps are enclosed by fuel assemblies in the reactor core, and a total of 77 irradiation channels with diameters of 1.59-12.7 cm are distributed throughout the reactor.

HFIR [27,28] was founded in 1961 and reached its first criticality on August 25, 1965. Full-power operation was achieved in September 1966. The initial motivation for constructing HFIR was to generate significant and weighable quantities of heavy elements (Cm, Bk, Cf, Es, Fm, etc.), particularly ² ²Cf, to support fundamental research and applications of transplutonium elements. Therefore, a research reactor with an extremely high neutron flux was required, which would accelerate the production efficiency of transplutonium nuclides, enable production of radioisotopes with high specific activity, and facilitate material tests. HFIR is a light-water-cooled and moderated HFR with a rated power of 125 MW. As shown in Fig. 5 [Figure 5: see original paper], the reactor core consists of a series of concentric fuel elements, and the fuel region is divided into inner and outer fuel zones. A flux trap is located at the center of the reactor core, where the maximum total neutron flux reaches 5×10^1 n/(cm²·s). providing basic conditions for transplutonium isotope production. In addition, various irradiation targets can be installed in the beryllium reflector, and four horizontal neutron beams can be used for neutron scattering and imaging.

HFETR [13,28] is a versatile HFR in China, designed as a vessel-type reactor with a design power of 125 MW. The reactor uses U Si -Al dispersion fuels and is cooled and moderated by light water. HFETR is characterized by high neutron flux, flexible reactor core layout, large irradiation space, and short irradiation periods. The maximum thermal neutron flux is 6.2×10^1 n/(cm²·s) (E < 0.625 eV), and the maximum fast neutron flux is 1.7×10^1 n/(cm²·s) (E > 0.625 eV). Construction of this reactor began in 1971, achieved first criticality in 1979, and started high-power operation in 1980. In 2007, the conversion of HFETR from HEU fuel to low enrichment uranium (LEU) fuel was completed [29], with ²³ U enrichment decreasing from 90 wt% to 19.75 wt%. Over the past



few decades, HFETR has continuously conducted irradiation research on fuel and materials for power reactors and radioisotope production for industrial and medical applications. This reactor is expected to remain operational until 2028.

SM-2 was reconstructed and updated to SM-3 [22] between 1991 and 1992. The physical startup of SM-3 was conducted in December 1992, and the power startup in April 1993. This reconstruction project set the lifespan at 25 years. SM-3 removed the horizontal channels and added additional vertical experimental channels, including one central neutron flux trap, six channels in the active core, and 30 channels in the reflector. The maximum thermal neutron flux in SM-3 is $5.0\times10^1~\text{n/(cm}^2\cdot\text{s)}$ (E < 0.625~eV), which occurs in the central flux trap of the reactor core, and the maximum fast neutron flux is approximately $2.0\times10^1~\text{n/(cm}^2\cdot\text{s)}$ (E > 0.1~MeV). This reactor is primarily used for transplutonium nuclide production (10–25 mg 2 ²Cf per year), nuclear fuel production, and material testing. From 2017 to 2020, a modernization update [30] of SM-3 was conducted, including replacement of reactor internal structures and digital control and protection systems, with an expected lifespan extension until 2040.

2.3 Current Stage

Since the beginning of the 21st century, the development of nuclear energy and technology in various countries has increased the demand for HFRs. Several innovative HFRs, such as OPAL and CARR, have been designed and constructed. In addition, several HFRs have been planned and are under construction for multipurpose applications.

CARR [14] is an HFR with satisfactory performance and multiple applications. This reactor reached its first criticality in May 2010 and entered full-power operation in 2012. CARR is a tank-in-pool reactor enclosed by a slightly pressurized aluminum container and submerged in the reactor pool. With this structure, the reactor core would not be exposed even under accident conditions. A cross-sectional view of the reactor core and surrounding reflector is shown in Fig. 6 [Figure 6: see original paper]. The nominal operating power of CARR is 60 MW, and the reactor core is loaded with 21 fuel assemblies using U Si -Al dispersed fuel plates. The reactor is light-water-cooled and moderated, and a heavy-water tank is used as a reflector. It contains 25 vertical and 9 horizontal irradiation channels, which are almost entirely located in the reflector. Only a few channels are arranged near the boundary of the aluminum container in the reactor core. Applications of CARR include fuel and material irradiation tests, radioisotope production, neutron scattering and imaging, and neutron activation analyses.

OPAL [15] is an open-pool HFR in Australia that achieved its first criticality on August 12, 2006. The nominal power is 20 MW, and it adopts low-enriched U Si-Al dispersion plate fuel with aluminum cladding. The reactor core is cooled and moderated using light water and surrounded by a heavy-water reflector. The maximum thermal neutron flux is $3.0\times10^1~\rm n/(cm^2\cdot s)$, which appears in the irradiation position in the reflector, and the maximum fast neutron flux is



 2.0×10^1 n/(cm²·s). The reactor pool is connected to the service pool through a transfer canal. The layout of the reactor core and irradiation channels in OPAL is shown in Fig. 7 [Figure 7: see original paper]. Since its establishment, OPAL has become an important supplier of medical and industrial radioisotopes (such as Mo, mTc, ¹³¹I, ¹³Sm, ¹Cr, etc.), a crucial testing platform for materials science and research, and a service provider for silicon transmutation doping.

The Jules-Horowitz Reactor (JHR) [32,33] is a new HFR under construction in France. The nominal design power of this vessel-in-pool reactor is 100 MW. The reactor is fueled by cylindrical U Si -Al assemblies. The reactor core is cooled using light water in a slightly pressurized primary circuit, and a beryllium reflector is used. The reactor core and reflector also provide sufficient irradiation positions for fuel and material irradiation tests, offering instrumented and circuit irradiation test capabilities. The expected fast neutron flux in the irradiation position is 5.5×10^1 n/(cm²·s) with energy above 1 MeV, corresponding to a radiation damage of 16 dpa per year. JHR also enables a high thermal neutron flux in the irradiation position, which can be used for radioisotope production. The reactor is planned to operate after 2030 and will enhance irradiation research capabilities for fuel and materials in France and Europe.

To support the development of advanced fast reactors [34] within the scope of fourth-generation nuclear energy systems, the USA, Russia, and China have planned to build new HFRs with fast neutron spectra. Most of these are designed as liquid-metal fast reactors [23]. The Multi-Purpose Fast Research Reactor (MBIR) [35], designed by Russia, is a loop-type versatile high flux fast neutron reactor with an asymmetric active zone layout. The maximum neutron flux of MBIR can reach 5.3×10^1 n/(cm²·s). Three experimental loops, three loop channels, and 14 material science assemblies are installed in the reactor. MBIR is designed to perform irradiation tests with Pb, Pb-Bi, Na, He, and molten salt, which can fulfill the development requirements of sodium-cooled and molten salt reactors. The designed lifespan of MBIR is 50 years, and it is expected to achieve its first criticality and begin operation by 2028. The Versatile Test Reactor (VTR) [24,36] in the USA is another high flux fast reactor with a power of 300 MW and a maximum neutron flux of 4.5×10^1 n/(cm²·s). High flux fast neutron reactors typically utilize sodium metal or lead-bismuth alloys as coolants. In China, a concept of ultra-high flux fast neutron reactor (UFFR) [37] has been proposed to achieve a maximum neutron flux at the 10¹ $n/(cm^2 \cdot s)$ level.

Recently, the Tsinghua High Flux Reactor (THFR) was designed by Tsinghua University, China. THFR is a light-water-cooled reactor with good safety characteristics. Its maximum neutron flux reaches $5.7\times10^1~\rm n/(cm^2\cdot s)$, and both its thermal and fast flux can achieve $2.0\times10^1~\rm n/(cm^2\cdot s)$, which is expected to provide a wide range of applications.

3 Design Features of the High Flux Reactor

HFRs possess technical features that distinguish them from power reactors and other reactor types. These features include not only a series of measures adopted to achieve sufficiently intensive neutron fluxes but also special technologies to improve irradiation capability, operational flexibility, and safety. This section analyzes the technical features of HFRs in terms of reactor core design, irradiation capability, and operating characteristics.

3.1.1 Reactor Physics and Thermal-Hydraulic Design

The fundamental design requirement of an HFR is to provide the necessary neutron flux conditions for irradiating samples at the lowest possible power and economic cost. To describe the law governing high neutron flux realization using a homogeneous reactor as an example, the total neutron flux can be described as:

$$\phi = \frac{q_V}{E_f N_f \sigma_f}$$

where q_V is the power density, E_f is the average heat release energy per fission, N_f is the atomic density of nuclear fuel, and σ_f is the effective one-group fission cross section.

Clearly, the neutron flux level can be improved by increasing the power density, reducing the fuel loading, and reducing the one-group fission cross section. Noting that the fission cross section is generally smaller in the fast energy range than in the thermal energy range, the total neutron flux is expected to improve if fission events are dominated by fast neutrons. However, moderated neutrons can contribute to reducing the critical fuel loading. These two contradictory aspects must be considered when designing HFRs.

Several HFRs adopt the design concept of inverse-neutron flux traps [38]. The reactor core is compact and undermoderated, and fast neutrons leak into the surrounding reflector area where they are moderated, resulting in a high thermal neutron flux in the reflector. The determination of the thermal neutron flux level depends on the objective of the study. In the example of ² Cm production through the transmutation of ² Pu, the "bottleneck" for transmutation is the transition from ² Pu to ² Pu. When the characteristic burnup time is set as $1/(\Phi\sigma_c) \leq 0.5$ year, the neutron flux should be $\Phi \geq 1/(\sigma_c \times 0.5 \text{ year}) = 3.4 \times 10^{15} \text{ n/(cm}^2 \cdot \text{s})$.

An increase in the power density q_V directly results in a higher neutron flux. For thermal-hydraulic design, the heat released from fuel assemblies represents a burden that significantly increases the cost of loop cooling. The main methods for increasing the allowable q_V are to reduce the coolant temperature at the core inlet, increase the coolant velocity, and expand the heat exchange surface by using fuel elements with thin shapes, such as plates.

The primary loop pressure of HFRs is significantly lower than that of power reactors. For example, the operating pressure of SM-3 is 4.9 MPa [22]. On one hand, relatively low pressure ensures that experimental setups operate under acceptable conditions; on the other hand, increasing pressure increases the subcooling margin, delays the critical boiling transition, and promotes the growth of q_V . In addition, a lower inlet temperature enables the coolant temperature to remain further away from the melting point of the fuel cladding. The compromise results in a subcooling margin of 70 to 200 °C and a coolant temperature in the primary loop that does not exceed 100 °C.

The coolant velocity is significantly higher than that of power reactors. For example, the maximum velocity in SM-3 is 13.5 m/s [22] and in HFIR is 15.5 m/s [11]. High coolant velocity enhances heat transfer between the coolant and cladding surface; however, it is primarily limited by flow-induced vibrations, thermal stresses on the cladding, and surface corrosion and erosion. The channels between plate-type fuels are generally narrow, and small disturbances during flow produce reciprocal movements at high flow rates. On one hand, this increases the risk of fuel meltdown due to deformation; on the other hand, alternating deformation increases fuel fatigue. Furthermore, increased coolant velocity results in more intensive dissolution of active substances in the coolant toward the surface and tears off part of the corrosion film from the surface, i.e., corrosion and erosion of the fuel cladding. Notably, due to the high heat flux density, the oxide layer on the aluminum cladding dominates the non-negligible temperature margin. The thickness of the oxide layer increases progressively with prolonged operation, limiting the reactor's operating time.

In summary, to determine the maximum q_V of the designed HFR, the following aspects must be considered: (1) fuel element structure and heat transfer parameters; (2) determination of heat transfer conditions and critical heat flux; (3) temperature stresses at high energy densities per volume of the fuel element; (4) stability at high coolant velocities; and (5) limitations imposed by corrosion and erosion on the fuel cladding.

3.1.2 Reactor Body Structure

The reactor body structure refers to the basic configuration of the reactor core, pressure vessel, reactor pool, and other reactor-affiliated facilities. The reactor body structure of an HFR can be roughly divided into three forms: pool, vessel (or tank), and vessel-in-pool (or tank-in-pool). The difference between these structures lies primarily in two aspects: whether the reactor core is enclosed in a pressure vessel (or other pressurized containers), and whether the reactor core vessel is submerged in a reactor pool. The pressurized vessel or tank aids in improving the reactor operating pressure, which enables larger reactor core power density and thus higher neutron flux. A reactor pool filled with water increases reactor safety and facilitates convenient operation with irradiation targets and other test samples.

The open pool reactor structure is widely used, as seen in MJTR and OPAL. For CARR, based on the pool reactor design scheme, the reactor core is contained in a slightly pressurized tank fabricated from aluminum. ATR and HFETR are vessel-type HFRs, whereas HFIR is a representative vessel-in-pool reactor, with operating pressure typically at 2–3 MPa. The basic structure of a vessel-in-pool reactor improves the neutron flux level with satisfactory safety features, making it preferable for ultra-high-flux reactors. The design concept of inverseneutron flux traps is typically adopted for pool-type HFRs [38]. The reactor core is compact and undermoderated, and fast neutrons leak into the surrounding reflector area where they are moderated, resulting in a high thermal neutron flux in the reflector.

A pressure vessel is an important device that serves as a second safety barrier to prevent the leakage of radioactive materials. It also provides support and positioning for the internal structure of the reactor. Pressure vessels are typically made of austenitic stainless steel (such as 316LN), which has higher toughness at various temperatures.

3.1.3 Fuel Assembly

HFR systems are simpler than power reactors and typically operate at lower temperatures. The uranium fuel loading of an HFR is significantly lower than that of a power reactor. An HFR typically requires only a few kilograms of nuclear fuel. For example, the total ²³ U loading in the HFIR core is 9.4 kg.

However, HFR fuel assemblies may require uranium with high enrichment of 20 wt% or higher. Historically, many HFRs used HEU fuels with enrichment higher than 90 wt%, such as HFIR, ATR, and SM-3. With HEU fuels, the absorption by ²³ U and other reactor core materials can be minimized, resulting in a more compact reactor core structure and longer operating period. In addition, the utilization of HEU can significantly reduce HFR costs, including fuel and construction expenses. Although the unit price of natural uranium is lower than that of HEU (by approximately one-third), reactors using natural uranium require much higher power to achieve the same neutron flux level as reactors using HEU, resulting in much higher fuel burnup. Furthermore, the cost of the first fuel load must be considered. Based on an analysis by Bath [2], the ²³ U contained in 1 t of natural uranium is about 30 times more expensive than 10 kg of HEU. Moreover, the cost per unit mass of ²³ U increases significantly when enrichment exceeds 90%. Therefore, the Soviet Union used 90 wt% HEU, whereas the United States used 93 wt% HEU.

Under the Reduced Enrichment for Research and Test Reactors (RERTR) program proposed in the 1970s [39], many HFRs were converted from HEU fuels to LEU fuels. In HFETR, a U Si -Al dispersion fuel with an enrichment of 20 wt% was used to replace the UAl alloy fuel with an enrichment of 90 wt%, and the thickness of the fuel meat was adjusted accordingly.

Due to the high burnup of HFRs, the fuel assembly should with stand radiation damage and accommodate fission products effectively. The investigations and wide utilization of dispersion fuels, such as U-Al or U Si [40], originate from research reactors including HFRs [2]. Dispersion fuel exhibits good compatibility with the aluminum matrix, high thermal conductivity, satisfactory capability to retain fission gases, and better fabricability.

Fuel cladding made of aluminum or zircaloy is adopted, which has high strength, a small neutron absorption cross-section, and better corrosion and radiation resistance performance. The U Si fuel has been used to substitute U-Al fuel under the RERTR project. To compensate for the decrease in 23 U enrichment, the uranium density in the fuel assembly must be increased. For innovative LEU fuels [5], such as U-Mo alloys, the uranium density can reach 10 g/cm³ or higher.

HFRs often have extremely high power densities in the reactor core, up to 1000 MW/m³ or higher. Plate or casing assemblies are widely used in almost all later HFRs [41] because of their better cooling performance (larger surface area-to-volume ratio), better mechanical and vibration stability (particularly the curved structure), and lower manufacturing difficulty (compared with cruciform fuels). The evolution of the plate fuel assembly in an HFR is shown in Fig. 8 [Figure 8: see original paper]. Fuel plates are typically manufactured using a rolling process, whereas fuel bundles are typically processed using a coextrusion approach.

3.1.4 Coolant and Moderator

Most HFRs use forced circulation cooling, with coolant flowing through the reactor core from top to bottom [42,43]. Commonly used coolants include water and liquid metals, and typical moderators are light water, heavy water, and beryllium.

A large proportion of HFRs use light water as both coolant and moderator. Due to the strong shielding effect of light water on neutrons, the average free path of neutrons in light water is relatively short. When neutrons travel from a fission event to the irradiation channel, a significant proportion of attenuation occurs in the neutron flux. In addition, the water layer in the thermal neutron channel has a strong shielding effect on fast neutrons. The average free path of neutrons in heavy water is longer than that in light water, which potentially contributes to a higher neutron flux.

Liquid sodium or lead-bismuth is often adopted as the coolant in fast-neutron HFRs, such as VTR and MBIR. The satisfactory thermal performance of liquid metal coolant guarantees efficient heat transfer from the fuel assemblies. Sodium-cooled HFRs typically require an intermediate loop, which also increases system complexity. The boiling point of lead-bismuth alloy is 1670 °C at atmospheric pressure, which potentially improves the reactor's inherent safety. The melting point of this alloy is relatively low (125 °C), enabling the reactor to operate at low temperatures and reducing requirements on structural materi-



als. In addition, no significant volume change occurs during the transformation from solid to liquid, which is favorable for frequent startup and shutdown of the reactor. For these reasons, newly proposed fast-neutron HFRs tend to adopt lead-bismuth alloy as a coolant [44]. However, some challenges in lead-bismuth-cooled reactors still exist, such as corrosion of fuel and structural materials.

3.1.5 Reflector

The presence of reflectors reduces the critical fuel mass and increases the neutron flux-to-power ratio. Common reflector materials include water, heavy water, beryllium, beryllium oxide, and graphite. Based on a single-group model for spherical reactors using different moderators and reflectors, Bath et al. [2] identified two configurations that provide the largest ratio of maximum neutron flux to reactor power: reactors using water moderator and beryllium reflector $(6.78\times10^1 \text{ n/(cm}^2\cdot\text{s}\cdot\text{kW}))$, and reactors using heavy water as both moderator and reflector $(7.34\times10^1 \text{ n/(cm}^2\cdot\text{s}\cdot\text{kW}))$.

Therefore, an HFR typically utilizes beryllium or heavy water as a reflector surrounding the reactor core to reduce neutron leakage. The neutron absorption cross-section of beryllium or heavy water is smaller than that of light water, which results in a higher thermal neutron flux and enables irradiation experiments in the HFR. The arrangement of irradiation channels in a heavy-water reflector is relatively convenient; however, adjusting the structure and position of the reflector is difficult. Moreover, the system structure related to heavy water is complex and requires higher maintenance costs, as well as sealing and cooling.

Beryllium also exhibits defects during long-term operations and shutdowns. It is prone to radiation swelling and embrittlement after being employed for decades [45] and experiences beryllium poisoning during shutdowns [46]. To ensure safe reactor operation, radiation supervision and regular replacement of beryllium components are necessary.

3.1.6 Reactivity Control Mechanism

Reactivity control is the basis for reactor startups, shutdowns, and power adjustments. The reactivity control elements of HFRs typically include control rods, control plates, and rotation control drums [47], as shown in Fig. 9 [Figure 9: see original paper]. The main distinctions between the various control elements are their action modes, such as linear or rotational movements. Most HFRs, such as HFIR and HFETR, adopt in-core control rods or control plates, whereas only a few HFRs, such as ATR, use rotational control drums. Some have adopted a control mechanism that combines a control drum and control rod, with the control drum serving as the main approach.

Compared with control rods in the core, control drums installed in the reflector can reduce neutron absorption and flatten the axial power distribution. However, the control worth is typically smaller than that of control rods. The main



technical parameters of the control drum include the rotation angle range (typically between 0° and 180°), rotation velocity, single-rotation limit, rotation position accuracy, quick reset time, temperature, pressure, and type of working medium.

A follower fuel assembly is typically adopted as the control element inserted into the reactor core. When the control element is withdrawn from the reactor core, the follower fuel element is inserted into the reactor core, which helps compensate for the reduced reactivity.

For more convenient operation of irradiation devices in the upper space of the reactor, the control drive motors of various control elements are typically located below the reactor core. CARR [14] has four compensation control rods in the reactor core and two safety rods in the heavy-water reflector adjacent to the reactor core. The driving mechanism of the compensation control rod is arranged at the bottom of the reactor core and adopts magnetic driving technology, while the safety rods use hydraulic drive technology with diverse design principles different from those of the compensation control rod, which helps avoid common cause failures and improve reactor safety.

3.1.7 Hydrochemistry

The hydrochemical requirements of HFRs are less complex than those of power reactors due to their lower coolant operating temperatures and shorter fuel lifetimes. An HFR is more concerned with maintaining a low level of radioactivity in the coolant than with scale formation [2]. This radioactivity primarily originates from reactive gases and radioisotopes dissolved in water. In addition to isotopes ¹ O and ¹Ar, fluorine and nitrogen isotopes ¹³N and ¹N also contribute to gas activity. Radioisotopes dissolved in water originate primarily from supplemental water and circuit materials. Sodium and calcium salts in the makeup water are irradiated to produce ²Na and Ca. Structural materials and fuel cladding enter the coolant due to corrosion after prolonged irradiation.

In addition, the coolant should be slightly acidic to minimize corrosion of aluminum cladding and beryllium.

According to Vladimirova [48], Russia proposed the following hydrochemical standards for the primary loop of research reactors: pH value at 25 °C, 5.0–6.5, $\pm 1\%$; mass concentration of chloride ions, not more than 50.0 µg/kg, $\pm 20\%$; mass concentration of aluminum, not more than 50.0 µg/kg, $\pm 5\%$; mass concentration of iron, not more than 50.0 µg/kg, $\pm 5\%$; total specific activity by fission products, not more than 2.5×10 Bq/L.

3.2.1 Irradiation Facilities

An HFR is primarily used for neutron irradiation. The capabilities and performance considered in irradiation tests primarily include neutron flux, neutron energy spectrum, irradiation space, irradiation time, temperature, pressure, gas



environment, and online monitoring. The irradiation facilities [49-51] typically include static irradiation devices, instrumented irradiation devices, irradiation test loops, and rabbit test devices.

A static irradiation device is used to enclose and fix irradiation test specimens, typically in the form of irradiation jars or tanks, frequently in an inert gas or water environment. A small irradiation jar can be arranged in the reactor core or reflector and can be designed with flexibility for insertion and uploading operations. Some samples with low heat generation are suitable for closed irradiation devices, whereas others may require coolant flow channels. Larger irradiation tanks are located outside the reactor core to utilize leaked neutrons.

Working conditions, such as the temperature of the static irradiation device, cannot be monitored during irradiation but only rely on reactor operation. In contrast, the instrumented irradiation device provides the capability to monitor online irradiation parameters such as irradiation temperature, gas composition, coolant temperature, coolant pressure, and coolant velocity. It is equipped with various monitoring devices, including thermocouples, neutron detectors, flow meters, sampling tubes, pressure tubes, gas tubes, and electric heating elements. The temperature of an irradiated sample is the most commonly monitored and controlled parameter. Ensuring that the temperature of the fuel sample does not exceed safety limits is particularly important. In addition, equipment for monitoring neutron fluence is fundamental for irradiation testing of advanced reactor materials [52].

As shown in Fig. 10 [Figure 10: see original paper], the irradiation test loop was designed to simulate the actual operating conditions of test specimens, typically utilized to irradiate nuclear fuel and materials with large heat generation. A typical design is a central neutron flux trap, which is utilized in the SM [22] and HFIR reactors [27]. Fig. 10(b) shows the small irradiation capsule utilized in HFIR for fuel irradiation tests; it is located in the flux trap irradiation position. The aim of designing this trap is to obtain a higher thermal neutron flux than that of the test loop in the active zone. Hence, these traps are often filled with light water [2]. The coolant of the test loop is isolated from the primary coolant, and irradiation parameters, including pressure, temperature, and velocity, are controlled separately by the loop. The design and construction of an irradiation test loop are relatively complex. The Russian MIR.M1 reactor [17] has 11 experimental circuits, including water, steam, and gas test loops, which provide irradiation test environments for various reactors.

The rabbit test device can automatically and flexibly accomplish sample transportation, irradiation, and measurement. The rabbit system is composed of multiple rabbit capsules driven by pneumatic or hydraulic transmission and controlled by automatic or manual approaches. This test facility enables irradiation samples to be inserted into or removed from the target area during reactor operation, making it suitable for producing radionuclides with short half-lives, neutron activation analysis (NAA), and nuclear data measurement.



3.2.2 Post Irradiation Examination

The facilities for post-irradiation examination of irradiated nuclear fuel and materials are primarily used for transportation and temporary storage, cutting and dismantling, non-destructive testing of irradiation devices and samples, and post-irradiation inspection of spent fuel and materials. These include appearance and size analysis, weight measurement, burnup analysis, composition and microstructure analysis of irradiated samples, accident condition simulation, study of mechanical and thermal performance, and special research performed according to user requirements. The post-irradiation facility [53,54] primarily includes a hot cell and radioisotope processing device.

The cluster of hot cells is composed of heavyweight concrete hot cells, lead-shielded hot cells, and shielded glove boxes. The hot cell is often adjacent to the HFR, with transportation channels and waterways connected to the reactor. An interface device is used to transport irradiated fuel or material from an external transportation container to a temporary storage room in a hot cell. Special lifting facilities, master-slave robots, and power robots are used to conduct post-irradiation inspections of different types and structures of fuel elements and structural materials.

Post-processing for radioisotope production involves irradiated target dissolution, radiochemical separation and purification, and radioactive waste transportation. Radioisotope processing facilities include hot cell lines and several glove boxes. Highly radioactive isotopes need to be processed in the hot cell, while relatively less radioactive isotopes can be handled in glove boxes.

3.3 Operating Characteristics

In an HFR, the operating power and period change frequently according to the requirements of specific experimental missions, leading to different operating characteristics and safety standards [52].

The physical startup or first criticality is an important milestone in HFR construction. Physical startup experiments can verify the theoretical design of the reactor and obtain basic physical parameters, such as control rod worth and reactivity temperature coefficient, for subsequent operation. For instance, ATR has a critical testing device, ATRC, which is a 1:1 replicate of the reactor core. With the development of reactor physics analysis methods, the startup of an HFR can be accurately predicted using numerical simulations, which may substitute for the construction of critical experimental devices. The physical startup of CARR was entirely based on theoretical results from physics calculation codes [56].

Beryllium blocks can produce high-energy neutrons through the (,n) reaction [57]. The strong background generated by the photoinduced neutron effect of beryllium is effective in overcoming blind spots when the HFR operates at high power for a period without requiring an additional neutron source to restart. However, the presence of photoexcited neutrons also increases the difficulty of



extrapolating fuel elements during the initial fuel-loading procedure for criticality. In addition, the effects of beryllium poisoning [46] must also be considered. Be produces Li and ³He with strong thermal neutron absorption cross sections in fission reactions with fast neutrons. ³T from Li decays to ³He during reactor shutdowns, enabling large accumulation of ³He. The longer the reactor has been in operation and the longer the shutdown, the larger the negative reactivity and the shorter the allowable shutdown time [58]. The beryllium status must be monitored prior to startup.

In an HFR, the surface heat flux of the fuel assembly is large. As mentioned previously, the operating pressure and temperature of the coolant are relatively low, but the coolant flow speed is considerably high to ensure sufficient cooling capability. Irradiation test components or targets affect the reactivity and power distribution of the HFR core [59] and cause significant heat generation. Therefore, additional requirements exist for cooling the experimental equipment [60]. Because the coolant channel is narrow, blockage accidents in the coolant channel must be prevented during operation [61,62].

Under high- or full-power operating conditions, the coolant of an HFR flows from top to bottom to cool the reactor core, driven by the primary pump. During low-power or shutdown operation, the reactor is cooled by natural circulation, and the coolant flows from bottom to top. This means that the coolant flow direction reverses, which results in zero coolant velocity during reversal, causing the convective heat transfer coefficient to decrease and resulting in deterioration of heat transfer. The flow reversal effect [43] should be considered during reactor shutdown.

Special attention should be given to various operations and experiments that may cause reactivity changes during operation. The reactivity disturbance value and rate should be accurately estimated. Irradiation test components should be considered part of the reactor, and their reactivities should be evaluated individually. Notably, when a large number of irradiated samples are placed in the irradiation channel, the neutron flux is significantly reduced due to large absorption, resulting in failure of the irradiation channel. The operation of large reactivity components should be performed during shutdown cooling periods, whereas the manipulation of small reactivity components can be performed without reactor shutdown.

The lifespan of a vessel-type HFR depends primarily on the life of its pressure vessel, which is determined by the irradiation effect of fast neutrons (energy typically higher than 0.1 MeV). Early designed HFRs opted for pressure vessel replacement, such as HFIR. The SM-3 reactor was constructed with a new pressure vessel inside the old one [22]. Therefore, monitoring the fast neutron fluence in pressure vessels is necessary during the lifespan of HFRs. HFETR uses SUS321 stainless-steel rods instead of aluminum rods in the outermost reflector to extend the life of the pressure vessel.



4 Applications of High Flux Reactors

According to statistical data from the IAEA website [8], the applications of operational research reactors are summarized in Table 3. Due to its stronger irradiation capability and more complete experimental facilities, the HFR typically has significant advantages over other research reactors in various application areas. As shown in Fig. 11 [Figure 11: see original paper], based on the demand and importance of irradiation missions, the HFR is primarily applied in nuclear fuel and material tests, radioisotope production for medical and industrial use, and neutron science research.

4.1.1 Fuel Irradiation Tests

To further satisfy requirements for improving the efficiency, reliability, and safety of nuclear power plants, innovative fuel assemblies are continuously being developed to increase fuel burnup, extend refueling cycles, and enhance safety margins and reliability.

The technical parameters considered in nuclear fuel irradiation primarily involve neutron flux [63,64], neutron energy spectrum, channel dimension, gas environment, online monitoring, and fuel failure rate. Fuel irradiation tests [65] are primarily conducted in irradiation loops, which are completely sealed and isolated from the reactor core to prevent radiation contamination even if the tested fuel assembly is damaged. Simulation of nuclear fuel behavior under transient and accident conditions can also be conducted in specific irradiation loops. During irradiation, fission power of the fuel assemblies and gamma radiation increase the reactor core temperature. Fuel irradiation samples containing more fissionable materials require higher cooling rates and larger experimental volumes, which should also receive more attention.

Existing HFRs used for fuel tests typically utilize thermal neutrons. Rod-bundle test fuel assemblies are adopted for irradiation tests of pressurized water-cooled reactor fuels. A 4×4 test assembly contains four guide tubes and 12 test fuel rods with different fuel densities, enrichments, and other parameters. The actual operating conditions of PWR fuel assemblies are simulated during irradiation tests. From 2012 to 2014, HTR-PM fuel pebbles were irradiated in HFRs for qualification tests, with a total irradiation time of 355 EFPD [66]. Five fuel pebbles were fixed in the irradiation capsule, which was placed in the sweep loop irradiation test facility, and the central pebble temperature was maintained at $1050\pm50~{\rm ^{\circ}C}$. Post-irradiation examination showed satisfactory performance of the HTR-PM fuel pebbles, with no particle failure. Micro fuel testing has also become an important application, such as that performed in HFIR [67]. The micro fuel sample can be encapsulated in a sealed target and placed in a basket through which primary coolant flows without requiring an independent test loop.

With the development of advanced nuclear reactors, the demand for fast reactor fuel tests is increasing, e.g., U-Zr alloy, U-Pu-Zr alloy, UPuN, and MOX fuels.



Therefore, an irradiation testing circuit suitable for fast reactor fuel irradiation should be constructed to satisfy requirements for special working fluids, neutron fluence, and radiation damage dose.

4.1.2 Material Irradiation Tests

To achieve better technical specifications and safety, advanced nuclear energy systems typically use innovative materials with better radiation or corrosion resistance and the ability to withstand higher operating temperatures [68-70]. The irradiation temperatures and radiation damage required for different advanced nuclear energy systems [71] are shown in Fig. 12 [Figure 12: see original paper].

The irradiation temperatures required for advanced nuclear reactors, particularly VHTR (650-1050 °C), GFR (550-900 °C), MSR (550-700 °C), and fusion reactor (300-1000 °C), are generally higher than those of conventional fission reactors. The required radiation damage doses are also much higher, such as those in fusion reactors (150-200 dpa), MSR (100-180 dpa), SFR (90-160 dpa), and LFR (50-130 dpa). To fulfill the irradiation requirements of materials used in these innovative reactors, both the temperature and fast neutron fluence of the HFR should be improved.

For example, ATR is primarily used for fuel and material irradiation tests, and multiple material irradiation tests have been conducted [72]. Instrumented lead experiments are used to perform irradiation tests on structural materials, cladding, and fuel pins. Six pressurized water loops installed in the flux traps can be used for irradiation of structural materials, cladding, tubing, and fuel assemblies. Materials and fuels for naval reactors, high-temperature gas-cooled reactors, and Magnox reactors have been irradiated using ATR [72]. The fuel assemblies used in the Soviet Union's nuclear thermal propulsion were tested in the IVG.1M and IGR reactors [73].

4.2 Radioisotope Production

Radioisotopes [74] have wide applications in industry, medicine, agriculture, and related fields. Most industrial and medical radioisotopes, particularly neutron-rich radioisotopes with high specific activities, are produced by HFRs worldwide. Medical radioisotopes [75] form an important foundation of nuclear medicine, with applications primarily in diagnostics (medical imaging) and therapy (radiation used to kill cancer cells or lesions). Some transplutonium isotopes such as ²³ Pu, ² IAm, ² Bk, ² Cf, etc., play an irreplaceable role in industry, non-destructive detection, aerospace heat sources, reactor startups, and scientific research. Several isotope-based radioactive sources are required in industrial and agricultural areas, including Co, ¹³ Cs, and ¹ Ir.

Radioisotopes are primarily produced through neutron-induced reactions (such as (n,), (n,), (n,p), and (n,f)) occurring in irradiation targets. Most reactions involve radiative capture and fission [74]. In the radiative capture method, the

product and target nuclides belong to the same element; therefore, chemical separation of the product nuclide from the target is frequently disabled, which reduces the specific activity of the product to some extent. Multiple types of radioisotopes can be obtained using the fission method, which facilitates acquisition of carrier-free radioisotopes with high specific activity.

The preparation of radioisotopes involves several continuous processes [74], including appropriate selection of target materials, target fabrication and encapsulation, irradiation in the reactor, transportation to the hot cell, target disassembly, radioisotope separation and purification, processing and source encapsulation, quality control, product packaging, and shipping. During the procedure of loading irradiation targets in the reactor core, irradiation positions and periods are flexibly chosen according to requirements for producing different types of radioisotopes based on neutron fluxes and spectra. Radioisotope preparation requires radiochemical post-processing of irradiation targets, and radiochemical processing must be conducted in hot cells with corresponding radiation protection requirements. Radioisotope products should also satisfy quality control requirements, and specific activity, chemical purity, and radiochemical purity are the three main parameters for assessing radioisotope product quality. For medical radioisotopes, the production process requires establishment of a QA/QC system and good manufacturing practices (GMP) requirements [76].

4.2.1 Rare Isotopes Production Transplutonium elements (such as Pu, Am, Cm, Bk, Cf, Es, and Fm) include plutonium and its subsequent homologous elements in the periodic table. These nuclides are all prepared through artificial nuclear reactions due to their limited production amounts and rather difficult production processes; they are also referred to as rare isotopes.

The radioisotopes produced in an HFR are closely related to the neutron flux level. The production of transplutonium nuclides [77-80], such as ² Bk, ² Cf, and ² ³Es, requires ultra-high thermal or resonant neutron flux. Industrial radioisotopes with long half-lives typically require long irradiation periods.

The production of transplutonium nuclides such as 2 ²Cf is complex and technically challenging [81]. The conversion chain of 2 ²Cf from 2 ²Pu, 2 ¹Am, 2 Cm, or other target nuclides involves a series of more than 10 neutron capture reactions. Intermediate nuclides such as 2 ²Am, 2 Cm, and 2 Cm have large fission cross-sections, and the thermal neutron absorption cross-sections of 2 Cm and 2 Cm are relatively small, which results in a very low yield of 2 ²Cf. Therefore, 2 ²Cf production requires higher thermal neutron flux (the average thermal neutron flux is typically larger than 1.5×10^1 n/(cm $^2 \cdot$ s)), and fission losses during irradiation should be controlled as low as possible. Currently, only the USA and Russia can produce transplutonium nuclides on an industrial scale. The maximum thermal neutron fluxes of HFIR and SM-3 are both larger than 5.0×10^1 n/(cm $^2 \cdot$ s), and the irradiation targets are located in the central neutron flux trap in the reactor core. 2 ²Cf production in the USA began in 1952 by irradiating 23 Pu in MTR, and large-scale production was initiated in 1966



after the establishment of HFIR and REDC at ORNL. A total of 78 production campaigns for transplutonium nuclides were performed until 2019, with accumulated production of 2 ²Cf reaching 10.2 g while preparing 2 Bk (1.2 g), 2 ³Es (39 mg), 2 Fm (15 pg; 1 pg = 10 12 g) and other nuclides [79]. SM-3 was used for 2 ²Cf production, adopting actinide oxide aluminum metal ceramic targets.

The preparation of transplutonium targets is one of the most important issues in transplutonium production. The flowchart is shown in Fig. 13 [Figure 13: see original paper]. Irradiation targets are typically aluminum-based materials (plutonium, americium, and curium) containing transplutonium oxides. This type of target aids in improving thermal conductivity and preventing sintering of transplutonium oxide ceramics. The main approach for large-scale ² ²Cf production depends on irradiating ² ²Pu or mixed americium/curium targets. Due to the large cross-sections of many target nuclei (such as ²³ Pu, ² ²Pu, ² ²Am, ² Cm, etc.), the target material density should be reduced to decrease the self-shielding effect of the target nucleus and heat released in the target during irradiation. The quality and safety of the fabrication and encapsulation process of the transplutonium target should be well controlled, which involves high-temperature and radiation-resistant target preparation technology and high-sealing welding technology for irradiation targets.

By analyzing the cross-sectional energy spectrum distribution of intermediate nuclides in the conversion chain of transplutonium nuclides, the neutron capture cross-sections of major nuclides such as americium/curium are relatively large in the resonance energy region, which can increase the probability and yield of neutron absorption by transplutonium nuclides. This means that transplutonium production amounts can be further optimized by adjusting the neutron spectrum of the irradiation target to increase the capture-to-fission ratio [81-83].

Transplutonium targets typically operate continuously in the reactor for nearly six to eight months. The irradiated targets are uploaded from the reactor core and transported to hot cells for radiochemical reprocessing. The post-processing primarily includes target dissolution, co-decontamination of transplutonium and lanthanide elements, separation of transplutonium and lanthanide elements, and transplutonium separation.

4.2.2 Medical Isotopes Production Nuclear medicine [84] is one of the most important applications of radiation and nuclear technology. In contrast to external radiation therapy, radioactive materials are transported into organs to be examined or treated based on the patient's metabolism. Radioisotopes for medical applications (such as Mo and ¹³¹I) form the basis of nuclear medicine and healthcare. Due to the short half-life of radioisotopes, nuclides are difficult to store over long terms.

High or ultra-high neutron flux increases the efficiency of producing medical radioisotopes with high specific activity in shorter times. The production of medical radioisotopes (such as Co, ³Ni, Sr, Y, and ¹ Lu) on an industrial

scale typically requires a neutron flux larger than $10^1~\rm n/(cm^2 \cdot s)$, and higher neutron flux contributes to achieving higher specific activities. Short-lived medical radioisotopes frequently require online processing of irradiation targets. A broader neutron energy spectrum provides more flexible selection for irradiating a radioisotope target with an appropriate neutron flux. The production of radioisotopes that are fission products typically involves two approaches: the fission-based method and the radiative capture method (or activation method). OPAL was one of the earliest HFRs to use an LEU target for Mo irradiation. The fission method facilitates acquisition of carrier-free medical radioisotopes with high specific activity. In addition, the radiative capture method is more sensitive to the neutron flux level; thus, HFRs have potential advantages for increasing specific activity through the activation method.

The HFRs used for medical isotope production and supply include HFR (Netherlands), BR2 (Belgium), SAFARI-1 (South Africa), and OPAL (Australia). The main radioisotope products are Mo/ mTc, ¹ Lu, Sr, and ¹³¹I. In the coming years, several major medical radioisotope production reactors worldwide will encounter issues such as aging, shutdown, and decommissioning. After 2030, only OPAL and FRM-II will be able to supply medical isotopes on a large scale. The production capacity gap for medical radioisotopes remains significant, with global shortages of Mo and ¹ Lu exceeding 10,000 Ci/week and 50,000 Ci/year, respectively. The global supply-demand contradiction will further intensify, and important medical radioisotopes will face shortage risks. To satisfy demands for radioisotope production, the Dutch government approved construction of a novel HFR, PALLAS [85], in 2012 to replace the operating HFR in Petten. The PALLAS reactor will have a thermal power of 55 MW with a design lifespan of 40 years and will be primarily used for producing medical radioisotopes. Construction has already begun.

The irradiation periods of several radioisotopes with short half-lives are shorter than the refueling period, such as Mo, whose production reaches an equilibrium value after an irradiation period of 5–7 days. Excess irradiation will not increase production amount and will increase the concentration of radioisotopes with long half-lives and high-level radioactivity. Therefore, online processing of the irradiation target is required, which involves pressure boundary sealing issues caused by the top opening of the pressure vessel. The quality control limitations of radioisotope targets include the maximum reactivity disturbance limit, cooling flow limit, and gamma heat release rate of the material.

4.3.1 Neutron Scattering

Neutron scattering technology [86] is based on measuring and analyzing changes in momentum and energy after interaction between thermal neutrons (or cold neutrons) and materials, and can be used to study the microstructure and dynamics of materials. Cold neutrons are neutrons with energy levels of meV or lower that are suitable for neutron scattering, neutron imaging, and other applications. Neutron scattering involves neutron diffraction, large-scale struc-



tural scattering, inelastic scattering, and other effective means [87,88]. Typical neutron scattering spectrometers include small-angle neutron scattering, time-of-flight polarized neutron reflection, and cold-neutron triple-axis spectrometers.

The quality of the neutron beam emitted from the reactor core is related to the position and angle of the neutron beam channel, which should be determined during HFR design. A neutron scattering spectrometer is selected based on the characteristics of the reactor core design. HFIR has four horizontal neutron beam tubes and 15 neutron scattering spectrometers [11].

Thermal neutron experimental spectrometers are typically arranged around the reactor core and utilize neutrons extracted from horizontal neutron channels. Thermal neutron scattering spectrometers include neutron stress analysis, thermal neutron triaxial, and neutron diffraction spectrometers [89], etc. The moderating effect of thermal neutrons increases the proportion of cold neutrons in the neutron energy spectrum. Cold neutrons with longer wavelengths are suitable for structural testing in the nanometer to submicron range, as well as for inelastic scattering measurements in the energy range below 10 meV. A mirror-structured guide tube can be used to guide neutrons over long distances to suitable spectrometers at appropriate positions. However, thermal neutrons cannot be released through a mirror-structured guide tube, which results in significant losses along the path. Ordinary cold neutron scattering spectrometers typically include small-angle neutron scattering spectrometers, time-of-flight polarized neutron spectrometers, and cold neutron triaxial spectrometers.

4.3.2 Neutron Imaging

For neutron imaging [90,91], the attenuation of intensity of a neutron beam passing through an object is used to perform perspective imaging of the measured object. Therefore, microstructure, spatial distribution, density, multiple defects, and related information on sample materials can be presented at a certain resolution.

Neutron imaging devices can be divided into fast neutron [92], thermal neutron [93], and cold neutron imaging devices [94]. Fast neutron imaging is suitable for testing large samples of various materials. The thermal neutron imaging device is located near the reactor core and has a high neutron flux, which is suitable for real-time imaging and large-scale sample testing. By selecting different filters and intercepting neutrons in different energy spectra, thermal neutron, epithermal neutron, and fast neutron imaging can be achieved. The cold neutron imaging facility in the cold neutron experiment hall has a low background and is suitable for application in high-resolution, symmetrical, and polarized neutron imaging. The non-destructive detection approach has significant technical advantages compared with X-rays and other traditional radiography technologies, making it suitable for imaging turbine blades, high-speed rail wheels, and nuclear fuel assemblies. In recent years, the combination of neutron imaging with digital technology and artificial intelligence has helped improve imaging



effects and achieve a wider range of applications [87].

4.3.3 Neutron Activation Analysis

NAA [95,96] is a non-destructive nuclear analysis method that uses a neutron beam with certain energy and current intensity to bombard a sample, resulting in activation of the measured element into a radioactive nuclide and measurement of the characteristic gamma-ray energy and intensity released by the generated nucleus. NAA technology has wide applications in physics, chemistry, materials, and energy. NAA can analyze multiple elements with high sensitivity, high accuracy, and low pollution, and is suitable for detecting samples in a wide range of sizes over shorter periods.

Compared with accelerator neutron sources and radioisotope neutron sources, the HFR neutron source has high neutron fluence, large activation cross-section for most elements, satisfactory spatial uniformity of neutron flux, and simple nuclear reaction pathway, which results in lower detection limits and higher sensitivity and accuracy. The reactor contains an NAA laboratory that analyzes samples irradiated in large pneumatic transport irradiation tubes for only a few minutes. They arrive at the laboratory within seconds from the reactor, and activation products with very short half-lives can be analyzed.

4.4 Other Applications

In addition to the major application fields described above, HFRs have extensive applications in other areas such as neutrino detection, silicon transmutation doping, and neutron therapy.

Neutrinos [97] are among the most fundamental particles in nature. They participate only in weak and gravitational interactions, have minimal interactions with matter, and exhibit extremely strong penetrating abilities. Neutrino research has always been at the forefront of particle physics research, which can help us understand the origin of matter and the universe and reveal the physical principles of the internal structure and evolution of celestial bodies, such as the sun and supernovae. The HFR has a compact reactor core with intensive neutrons, which is suitable for detecting ultrashort baseline neutrino oscillations and accurately measuring reactor-based antineutrino spectra while providing support for measurement of neutrino mass sequence [98-100]. In addition, performing basic physics research on neutrinos helps in understanding the nuclear reaction process inside fission reactors. PROSPECT [99] is a neutrino experiment facility installed in HFIR that is used to accurately measure the antineutrino spectrum of highly enriched uranium-fueled reactors and detect sterile neutrinos at an energy level of eV by searching for neutrino oscillations on a few-meter-long baseline.

Silicon transmutation doping [101-103] is widely used to dope silicon with phosphorus through the neutron capture reaction of 3 Si(n,) 3 1Si with subsequent

beta decay to ³¹P. Irradiated silicon is used to produce high-quality semiconductors. Silicon doping primarily utilizes the thermal neutron flux, and the fast neutron and gamma fluxes should be as low as possible. Silicon doping does not have very high requirements for neutron flux levels, and the irradiation facility is simpler; however, a larger irradiation space is required to accommodate a silicon ingot with a diameter of 8–12 inches. Therefore, this type of irradiation is suitable for pool-type HFRs, and a silicon ingot can be placed in the reflector surrounding the reactor core. Furthermore, the uniformity of irradiation and neutron spectrum screens should be considered in the design of silicon-doping schemes.

Reactor-based neutron therapy [104] is an important medical treatment approach, and the physical basis is neutron absorption reactions, such as 1 B(n,) Li, Li(n,) 3 H, or 1 Gd(n,) 1 Gd. The radiation released through these reactions is typically in the MeV range and results in high linear energy transfer (LET). The resulting biological effects are concentrated in the treated tissues. Boron neutron capture therapy (BNCT) [105] utilizes a thermal neutron beam to treat shallow tumors. Such facilities are typically located near hospitals. For the HFR, building a comprehensive therapy base for neutron therapy is appropriate. However, radiation protection and more flexible arrangement should be considered.

5 Future Trends

To satisfy future research and development requirements for advanced nuclear power systems, newly designed HFRs are pursuing better performance in the following aspects:

1. Higher Technical Parameters

The design of a novel HFR frequently requires higher technical parameters, including higher neutron flux, wider energy spectrum range, and better irradiation performance. For fast neutron spectrum irradiation, the fast neutron flux should typically lie in the range of 10^{12} to 10^{1} n/(cm²·s) with a radiation damage of 20-30 dpa per year. To achieve higher neutron flux, the power density of the reactor core should be increased as much as possible at an appropriate power level, typically adopting a compact reactor core structure. Additionally, the structural scheme of the reactor body, fuel assembly scheme, control approach, and coolant system should be determined. A broader neutron spectrum range is beneficial for various experiments such as radioisotope production and material irradiation tests [106]. The neutron spectrum of the irradiation device could be optimized by arranging neutron flux traps in the reactor core, selecting irradiation channel structures, and reasonably absorbing materials to improve neutron utilization in irradiation experiments. Newly built HFRs are typically equipped with multiple irradiation test loops with different irradiation environments for material tests and improved online information collection capability of the irradiation device with accurate detection information on



temperature, pressure, and fission gas concentration to further enhance HFR irradiation capability.

2. Higher Nuclear Safety Characteristics

While pursuing higher technical specifications, the safety goals for building new HFRs are constantly improving, and higher nuclear safety standards must be satisfied to ensure reactor safety under normal operating and accident conditions. The reactor should be designed with satisfactory inherent safety, which typically presents as a large negative reactivity temperature coefficient and strong self-protection capability. Innovative HFRs typically adopt passive safety systems and engineered safety facilities (primary pressure relief, safety injection, boron injection, containment isolation, and residual heat removal systems) to further improve reactor safety and mitigate accident consequences. The automation level of newly built HFRs should be appropriately improved to minimize reliance on operators and avoid the impact of human factors on HFR safety.

3. Comprehensive Versatility

The comprehensive use of irradiation resources is an important challenge for the application of new HFRs and is also an optimal way to fully utilize HFR irradiation resources. HFRs often prioritize one or more functions when considering other applications. For example, in HFRs dedicated to radioisotope production, targets are always arranged in irradiation channels with higher thermal neutron fluxes. The application of material irradiation tests and neutron beam extraction can also be considered as supplementary schemes. This design approach is beneficial for improving neutron utilization efficiency. The design of an innovative HFR is typically recommended to satisfy requirements for comprehensive versatility.

4. Application of Innovative Technology

The design and construction of an HFR is a complex system engineering project with high technical difficulty and a relatively long development cycle, involving a combination of multiple disciplines such as nuclear physics, fluid mechanics, thermodynamics, structural mechanics, and control technology. The use of several new technologies, such as digital technology, can improve the efficiency of design and engineering construction [38]. Digital systems adopt technologies such as digital twins and model-based systems engineering (MBSE) to build a digital design platform, digital engineering construction and debugging platform, digital delivery platform, and intelligent operation platform, providing efficient digital support for HFR development.

The following suggestions are proposed for future development, design, and application of HFRs:

1. The main challenge in developing HFRs is improving neutron flux while ensuring safety. A multipurpose HFR with comprehensive utilization of irradiation resources is one of the development directions. The simulta-

neous development of various HFR types is recommended. For example, a water-cooled high flux thermal neutron reactor is suitable for producing radioisotopes, particularly rare nuclides, and can achieve ultra-high thermal neutron flux with a mature nuclear engineering foundation. To address increasing demands for radiation damage to nuclear materials, a high flux fast-spectrum reactor is required, such as a sodium-cooled or lead-bismuth-cooled reactor.

- 2. In HFR design, the relationship between reactor core design and primary technical parameters (such as neutron flux, radiation damage, and operation period) should be considered. The design procedure depends on application requirements, which is suitable for establishing a goal-oriented design system. Reactor design should achieve optimal parameters for specific applications. A high utilization rate in HFR operation is also recommended.
- 3. To satisfy irradiation requirements of advanced nuclear fuel and materials, newly built systems should fulfill special irradiation parameter requirements (such as high neutron flux, sufficient irradiation volume, and sufficient irradiation damage) with specialized irradiation devices. The irradiation test facility is recommended to operate with special media such as liquid metal, molten salt, and supercritical water. The irradiation experiment should be capable of online monitoring of irradiation test parameters with high reliability. The HFR should also be equipped with advanced post-processing facilities to produce radioisotope products or perform fuel and material post-examinations. Irradiation test technology should match the reactor structure and specific irradiation test forms, including specimen partition layout, irradiation device design, irradiation parameter monitoring and control, and comprehensive irradiation environment control technologies. The irradiation test design should fully satisfy irradiation index requirements of specimens and ensure safety of irradiation tests and reactor operation. Due to demands for new irradiation tests, development of irradiation test technologies with a wider range of applications, more measurement parameters, and higher precision indicator control capabilities should be accelerated.

Acknowledgements

The authors would like to thank Dr. ZHANG Yebing for his constructive suggestions for improving this manuscript.

References

[1] IAEA, Research reactors: purpose and future. International Atomic Energy Agency (2016)

[2] G.A. Bat, A.S. Kochenov, L.P. Kabanov, Issledovatel' skie qdernje reaktorj: Uchebnoe posobie Issledovatel' skie qdernje reaktorj. Uchebnoe posobie dlq vuzov (Research Nuclear Reactors: Textbook for universities). (Atomizdat, 1972) (in Russian)

[3] S. O' Kelly, An Introduction to Research Reactors, in Encyclopedia of

- Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 1-7. doi:https://doi.org/10.1016/B978-0-12-819725-7.00054-4
- [4] N.P. Barradas, Applications of Research Reactors, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 8-25. doi:https://doi.org/10.1016/B978-0-12-819725-7.00110-0
- [5] IAEA, Commercial Products and Services of Research Reactors. Vienna: INTERNATIONAL ATOMIC ENERGY AGENCY (2013)
- [6] J.A. Lane, Ultra high flux research reactors. pp. Medium: ED. United States (1958), doi:10.2172/4301543
- [7] Y.G. Dragunov, I.T. Tretiyakov, A.V. Lopatkin et al., Modernization of the IBR-2 research pulsed reactor. At. Energy 112, 259-267 (2012). https://doi.org/10.1007/s10512-012-9591-9
- [8] Research Reactor Database (International Atomic Energy Agency, 2024). https://nucleus.iaea.org/rrdb/#/home.
- [9] H.F. McFarlane, Historical Survey of Test Reactor Programs at INL Over 70 Years, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 26-38. doi:https://doi.org/10.1016/B978-0-12-819725-7.00041-6
- [10] V.A. Tsykanov, A.V. Klinov, V.A. Starkov et al., SM reactor core modification solving materials-science problems. At. Energy 93, 713–717 (2002). https://doi.org/10.1023/A:1021763815423
- [11] J. Lane, R. Cheverton, G. Claiborne, et al., High Flux Isotope Reactor Preliminary Design Study. Oak Ridge National Lab.(ORNL), Oak Ridge, TN (United States) (1959)
- [12] D.S. O' Kelly, The Advanced Test Reactor, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 39-46. doi:https://doi.org/10.1016/B978-0-12-819725-7.00032-5
- [13] Z.Y. Zhang, Current status and application of the characteristics of HFETR. Nuclear Power Engineering 5(S2), 44-50+97 (1985). doi:CNKI:SUN:HDLG.0.1985-02-004 (in Chinese)
- [14] L.Z. Yuan, G.T. Ke, J.H. Jin, et al., Design Features and Innovative Technologies of China Advanced Research Reactor (CARR). Nuclear Power Engineering S2, 1-5. (2006). doi:CNKI:SUN:HDLG.0.2006-S2-001 (in Chinese) [15] D. Vittorio, M. Summerfield, The OPAL Research Reactor, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 147-154. doi:https://doi.org/10.1016/B978-0-12-819725-7.00096-9
- [16] S. Van Dyck, The BR2 Reactor: Short History, Characteristics and Utilization, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 96-104. doi.org/10.1016/B978-0-12-819725-7.00078-7
- [17] RIAR. Loop-Type Reactor MIR, https://niiar.ru/eng/node/226 [Accessed 12 April 2024].
- [18] G.A. Sun, D. Liu, J. Gong, et al., The neutron scattering platform of China Mianyang Research Reactor (CMRR) and recent applications. Science Sinica Physica, Mechanica & Astronomica 51(9) (2021). doi: 10.1360/SSPMA-2021-0096 (in Chinese)
- [19] G. Haas, The high flux reactor (HFR) nuclear research at NRG, In IAEA



- Consultancy Meeting «Catalogue of Research Reactors». Vienna, 2013
- [20] B. Piwowarski, M. Gryziński, R. Prokopowicz, et al., MARIA Research Reactor, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 124-131. doi:https://doi.org/10.1016/B978-0-12-819725-7.00138-0
- [21] H. Gerstenberg, W. Petry, FRM II (20 MW Germany), in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 113-123. doi:https://doi.org/10.1016/B978-0-12-819725-7.00139-2
- [22] V.A. Gremyachkin, Obosnovaniq osnovnjx inzhenerno-texnicheskix resheniy dlq povjsheniq akspluatacionnjx xarakteristik i bezopasnosti reaktora SM (Justifications of Basic Engineering and Technical Solutions for Improving Operational Characteristics and Safety of the SM Reactor), JSC RIAR, 2007. (in Russian)
- [23] G.R. Imel, Historical Survey of Fast Test Reactors, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 39-46. doi:https://doi.org/10.1016/B978-0-12-819725-7.00049-0
- [24] J. Roglans-Ribas, K. Pasamehmetoglu, T.J. O' Connor, The versatile test reactor project: mission, requirements, and description. Nucl. Sci. Eng. 196, 1–10 (2022). https://doi.org/10.1080/00295639.2022.2035183
- [25] IAEA, Research Reactor Core Conversion from the Use of Highly Enriched Uranium Fuels: Guidebook. (INTERNATIONAL ATOMIC ENERGY AGENCY, Vienna, 1980)
- [26] M.D. DeHart, Z. Karriem, M.A. Pope et al., Fuel element design and analysis for potential LEU conversion of the Advanced Test Reactor. Prog. Nucl. Energy 104, 117-135 (2018). https://doi.org/10.1016/j.pnucene.2017.09.007
- [27] D. Chandler, C.D. Bryan, High Flux Isotope Reactor (HFIR), Reference Module in Earth Systems and Environmental Sciences (2020). doi:https://doi.org/10.1016/B978-0-12-819725-7.00051-9
- [28] F. Simonen, A review of proposed upgrades to the high flux isotope reactor and potential impacts to reactor vessel integrity.
- [29] S.H. Sun, Y.X. Bu, T.Y. Duan, et al., HFETR low enrichment study, in Proceedings of the 30th Anniversary of the Operation of the High Flux Engineering Test Reactor (HFETR) (1980~2010), (China Academy of Nuclear Power Research and Design, Sichuan, 2007), pp. 336-339 (in Chinese)
- [30] A.L. Izhutov, PROGRAMMA RAZVITIQ Unikal' noy nauchnoy ustanovki «Vjsokopotochnjy issledovatel' skiy reaktor SM-3» (UNU SM-3) na 2020-2025 gg. (PROGRAM for the Development of the Unique Scientific Facility "High Flow Research Reactor SM-3" (HFR SM-3) for 2020-2025) (2020) (in Russian) [31] J. Schulz, ANSTO & OPAL Research Reactor Overview.
- [32] G. Bignan, The Jules Horowitz Reactor, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 132-140. doi:https://doi.org/10.1016/B978-0-12-819725-7.00098-2
- [33] D. Iracane, P. Chaix, A. Alamo, Jules Horowitz Reactor: a high performance material testing reactor. Comptes Rendus Phys. 9, 445–456 (2008). https://doi.org/10.1016/j.crhy.2007.11.003
- [34] I. Pioro, Handbook of generation-IV nuclear reactors. J. Nucl. Eng.



- Radiat. Sci. 3, 026501 (2017). https://doi.org/10.1115/1.4035327
- [35] IRC. MBIR reactor, https://en.mbir-rosatom.ru/reactor/ [Accessed 12 April 2024].
- [36] F. Heidet, J. Roglans-Ribas, Versatile test reactor conceptual core design. Nucl. Sci. Eng. 196, 23–37 (2022). https://doi.org/10.1080/00295639.2022.2091907 [37] W. Liao, F.X. Xia, H.X. Yu, et al., Requirement Analysis on Ultra-high Flux Fast Neutron Research Reactors. Nuclear Power Engineering 43, 222–226 (2022). doi:10.13832/j.jnpe.2022.06.0222 (in Chinese)
- [38] I.A.E. Agency, Utilization related design features of research reactors: a compendium. (International Atomic Energy Agency, 2009)
- [39] D. Wachs, RERTR Fuel Development and Qualification Plan. Idaho National Lab.(INL), Idaho Falls, ID (United States) (2007)
- [40] D.D. Keiser, D.M. Wachs, A.B. Robinson et al., The use of U Si /Al dispersion fuel for high power research reactors. J. Nucl. Mater. 528, 151820 (2020). https://doi.org/10.1016/j.jnucmat.2019.151820
- [41] X.L. Yin, G.L. Chen, H. Qian et al., A distributed parallel computing-based CFD analysis of plate-type fuel assemblies in CARR reactors. Int. Commun. Heat Mass Transf. (2024). https://doi.org/10.1016/j.icheatmasstransfer.2024.107335 [42] T.C. Liu, C.J. Yang, X.M. Liu, et al., Safe Design of Advanced Research Reactors in China, Nuclear Power Engineering 5(S2), 29–31. (2006) (in Chinese)
- [43] T.Z. Xu, T.Y. Duan, C.Y. Deng, et al., Analysis and Research of Flow Reverse in HFETR. Atomic Energy Sci. Technol. 45, 682–687 (2011). doi:10.7538/yzk.2011.45.06.0682 (in Chinese)
- [44] W.J. Li, F.X. Xia, H.X. Yu, et al., Conceptual Design Study of Ultra-high Flux Fast Neutron Research Reactor Fuel. Nuclear Power Engineering 43, 217–221 (2022). doi:10.13832/j.jnpe.2022.06.0217 (in Chinese)
- [45] M.Q. Feng, Y.H. Wang, J.T. Gu, et al., Post Irradiation Examination and Evaluation of Properties of Beryllium Assemblies in HFETR. pp. 8. China (2003) (in Chinese)
- [46] M. Wróblewska, D. Blanchet, A. Lyoussi et al., A review and analysis of the state of the art on beryllium poisoning in research reactors. Ann. Nucl. Energy 163, 108540 (2021). https://doi.org/10.1016/j.anucene.2021.108540
- [47] X.F. Wu, S. Li, C.H. Nie, et al., Buffer Test and Analysis of CENTER HFETR Control Rod Driver Mechanism. Nuclear Power Engineering. 38, 62–66 (2017). doi:10.13832/j.jnpe.2017.05.0062 (in Chinese)
- [48] O.N. Vladimirova, Vodno-ximicheskiy rezhim vjsokopotochnjx issledovateľ skix reaktorov MIR i SM (Water-chemical regime of high-flow research reactors MIR and SM), 2006 (in Russian)
- [49] IAEA, In-Pile Testing and Instrumentation for Development of Generation-IV Fuels and Materials., In International Atomic Energy Agency, Nuclear Fuel Cycle and Materials Section, Vienna (Austria), Dec 15, 2013
- [50] W.H. Yang, G.Z. Zhao, L. Zhang, et al., Irradiation Testing Capabilities and Irradiation Testing Technology of the HFETR. Nuclear Science and Engineering 38, 986–994 (2018) (in Chinese)
- [51] R.D. Li, X. Yang, G.B. Wang et al., An in-pile experimental loop for the



- irradiation of tritium breeding ceramics in China Mianyang research reactor (CMRR). Fusion Eng. Des. (2021). https://doi.org/10.1016/j.fusengdes.2020.112192 [52] G.X. Li, B.X. Zhou, M. Xiao et al., Overall development strategy of China's new-generation nuclear fuel. Strateg. Study CAE 21, 6-11 (2019).
- [53] P. Zhang, M.Z. Ran, Destructive Post-Irradiation Examination of Irradiated Fuel Rod. 39(S1), 113-116 (2005). doi:10.7538/yzk.2005.39.S1.0113 (in Chinese)
- [54] J.I. Cole, H. Ozaltun, J.-F. Jue, et al., Irradiation Performance: Research Reactor Fuels, in Encyclopedia of Nuclear Energy, ed. by E. Greenspan. (Elsevier, Oxford, 2021), pp. 420-428. doi:https://doi.org/10.1016/B978-0-12-819725-7.00140-9
- [55] Z. Lv, S.G. Xiao, W.Q. Zheng et al., The first critical experiment of China advanced research reactor. At. Energy Sci. Technol. 46, 1181-1184 (2012).
- [56] Z. Lu, Z.Y. Sun, S.G. Xiao et al., Application of MCNP code in the calculation of CARR nuclear commissioning. Sci. Technol. Rev. 30, 44-47 (2012).
- [57] S.H. Sun, L. Zhu, H.T. Li, Research on Nuclear Properties of Beryllium Reflector In Physical Startup of Reactor. Nuclear Power Engineering 36, 14-17 (2015). doi:10.13832/j.jnpe.2015.06.0014 (in Chinese)
- [58] A.P. Malkov, V.V. Pimenov, V.V. Kalygin, et al., Qderno-fizicheskie processj v berillii pod oblucheniem i ix vliqnie na fizicheskie i texnologicheskie xarakteristiki issledovatel' skix reaktorov (Nuclear-physical processes in beryllium under irradiation and their influence on physical and technological characteristics of research reactors), Proceedings of JSC SSC RIAR 13-25, (2016)
- [59] A. Shaimerdenov, D. Sairanbayev, S. Gizatulin et al., WWR-K reactor LEU core design optimization for improving the experimental characteristics. Ann. Nucl. Energy 195, 110174 (2024). https://doi.org/10.1016/j.anucene.2023.110174
- [60] F. Shen, Research on nuclear heating and heat transfer of irradiated targets in process of in-pile irradiation. At. Energy Sci. Technol. 37, 345-348 (2003).
- [61] K. Andrew, M. Kyle. Severe Accident Analysis of the Advanced Test Reactor. Proc., October 29-November 2 (2017)
- [62] Q. Lu, S.Z. Qiu, G.H. Su, Flow blockage analysis of a channel in a typical material testing reactor core. Nucl. Eng. Des. 239, 1978-1985 (2009). https://doi.org/10.1016/j.nucengdes.2008.06.016
- [63] L. Zhu, S.H. Sun, Y.X. Xiang et al., Study of neutron flux calculation for material irradiation in HFETR. Fusion Eng. Des. 125, 641–646 (2017). https://doi.org/10.1016/j.fusengdes.2017.05.062
- [64] T.Z. Xu, Y.X. Xiang, J. Li et al., Study on local construction of the 14MeV neutron field by LiD converter in HFETR. Fusion Eng. Des. 121, 288–295 (2017). https://doi.org/10.1016/j.fusengdes.2017.05.088
- [65] Y.Y. Zhang, W.H. Yang, L.Q. Fu et al., Study on thermal hydraulic characteristics of rod-type fuel irradiation test section. Proceedings of the 2023 Water Reactor Fuel Performance Meeting. (2024). Singapore: Springer Nature Singapore, 140–151 (2024).
- [66] S. Knol, S. de Groot, R.V. Salama et al., HTR-PM fuel pebble irradiation

- qualification in the high flux reactor in Petten. Nucl. Eng. Des. 329, 82–88 (2018). https://doi.org/10.1016/j.nucengdes.2017.09.020
- [67] T.D. Burchell, J.L. McDuffee, K.R. Thoms. Small Specimen Data from a High Temperature HFIR Irradiation Experiment. Paper presented at the Conference: ASTM STP1578: Symposium on Graphite Testing for Nuclear Applications: The Significance of Test Specimen Volume and Geometry and the Statistical Significance of Test Specimen Population, Seattle, WA, USA, 20130919, 20130920, United States, (2014)
- [68] IAEA, Research Reactor Application for Materials under High Neutron Fluence. (INTERNATIONAL ATOMIC ENERGY AGENCY, Vienna, 2011)
- [69] M.A. Fütterer, E. D' Agata, M. Laurie et al., Next generation fuel irradiation capability in the High Flux Reactor Petten. J. Nucl. Mater. 392, 184–191 (2009). https://doi.org/10.1016/j.jnucmat.2009.03.030
- [70] IAEA, Applications of Research Reactors towards Research on Materials for Nuclear Fusion Technology. (INTERNATIONAL ATOMIC ENERGY AGENCY, Vienna, 2013)
- [71] S. Zinkle, Structural materials for innovative nuclear systems (SMINS): workshop proceedings, Karlsruhe, Germany, 2007
- [72] C.J. Stanley, F.M. Marshall, Advanced test reactor: a national scientific user facility. 16th International Conference on Nuclear Engineering May 11-15, 2008, Orlando, Florida, USA. (2009), 367-372. https://doi.org/10.1115/ICONE16-48154
- [73] . Koroteev, G. Konyukhov, Y. Demyanko, Qdernje raketnje dvigateli (Nuclear Rocket Engines). (Norma-Inform, Moscow, 2001). (in Russian)
- [74] IAEA, Manual for Reactor Produced Radioisotopes. (INTERNATIONAL ATOMIC ENERGY AGENCY, Vienna, 2003)
- [75] S.M. Peng, Y.C. Yang, X. Xie, et al., Status and Prospect of Production and Application of Heap-illuminated Medical Isotopes in China. Chinese Science Bulletin 65, 3526–3537 (2020). doi:https://doi.org/10.1360/TB-2020-0374 (in Chinese)
- [76] W.H. Organization, WHO good manufacturing practices for pharmaceutical products: main principles, WHO Expert Committee on Specifications for Pharmaceutical Preparations. WHO Technical Report Series, No. 986, Annex 2. (2014) Geneva: World Health Organization. https://www.who.int/publications/m/item/trs986-annex2
- [77] J. Werner, W. Bickford, D.B. Lord et al., Summary of plutonium-238 production alternatives analysis final report. No. INL/EXT-13-28846. Idaho National Lab.(INL), (2013). doi: https://doi.org/10.2172/1091349
- [78] R.C. Martin, J.B. Knauer, P.A. Balo, Production, distribution and applications of californium-252 neutron sources. Appl. Radiat. Isot. 53, 785–792 (2000). https://doi.org/10.1016/S0969-8043(00)00214-1
- [79] S.M. Robinson, D.E. Benker, E.D. Collins et al., Production of Cf-252 and other transplutonium isotopes at oak ridge national laboratory. Radiochim. Acta 108, 737–746 (2020). https://doi.org/10.1515/ract-2020-0008
- [80] A.L. He, H. He, Z.Y. Li, et al., Key Technologies and Prospects for ² ²Cf Production. Isotopes Journal (2022). doi:10.7538/tws.2022.youxian.069 (in



Chinese)

- [81] W. Xu, J. Li, J. Zhao et al., An optimization design study of producing transuranic nuclides in high flux reactor. Nucl. Eng. Technol. 55, 2723–2733 (2023). https://doi.org/10.1016/j.net.2023.04.023
- [82] J. Li, J. Zhao, Z.H. Liu et al., Conceptual design study on Plutonium-238 production in a multi-purpose high flux reactor. Nucl. Eng. Technol. 56, 147–159 (2024). https://doi.org/10.1016/j.net.2023.09.019
- [83] M.Y. Wu, S.X. Wang, The investigation and calculation of the transmutation paths for the production of 2 $^2\mathrm{Cf}$ in fast reactors. Ann. Nucl. Energy 136, 107006 (2020). https://doi.org/10.1016/j.anucene.2019.107006
- [84] X.L. Lan, X.M. Song, Radiotheranostics: Challenges and opportunities. China Medical Imaging Technology 40, 3-6 (2024). doi:10.13929/j.issn.1003-3289.2024.01.001 (in Chinese)
- [85] W.J. Nuttall, P. Storey, Technology and policy issues relating to future developments in research and radioisotope production reactors. Prog. Nucl. Energy 77, 201–213 (2014). https://doi.org/10.1016/j.pnucene.2014.05.007
- [86] S.W. Lovesey, Theory of neutron scattering from condensed matter. (Clarendon Press Oxford, United Kingdom, 1984)
- [87] T.F. Li, M.M. Wu, X.S. Jiao, et al., Current Status and Future Prospect of Neutron Facilities at China Advanced Research Reactor. Atomic Nuclear Physics Review 37(03), 364–376 (2020). doi:10.11804/NuclPhysRev.37.2019CNPC36 (in Chinese)
- [88] H.Y. Huo, H. Li, Y. Wu et al., Development of cold neutron radiography facility (CNRF) based on China Mianyang research reactor (CMRR). Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. 953, 163063 (2020). https://doi.org/10.1016/j.nima.2019.163063
- [89] G.J. Zhu, M.J. Li, X.L. Liu et al., The latest progress on neutron texture diffractometer at China Advanced Research Reactor. Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. 1047, 167729 (2023). https://doi.org/10.1016/j.nima.2022.167729
- [90] IAEA, Neutron Imaging: A Non-destructive Tool for Materials Testing. (INTERNATIONAL ATOMIC ENERGY AGENCY, Vienna, 2008)
- [91] M.M. Wu, L.F. He, S.H. Ruan, et al., Application of Neutron Imaging Technology in Elemental Analysis. Chinese Inorganic Analytical Chemistry 1-15 (2024). doi:doi:10.3969/j.issn.2095-1035.2024.06.003 (in Chinese)
- [92] S. Wang, H.Y. Huo, W. Yin et al., Prototype design and experimental study of fission neutron imaging at China Mianyang research reactor. Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. 1061, 169061 (2024). https://doi.org/10.1016/j.nima.2023.169061
- [93] L.X. Zhang, S.Z. Chen, Z.D. Zhang et al., Resolution analysis of thermal neutron radiography based on accelerator-driven compact neutron source. Nucl. Sci. Tech. 34, 76 (2023). https://doi.org/10.1007/s41365-023-01227-x
- [94] E.H. Lehmann, A. Kaestner, L. Josic et al., Imaging with cold neutrons. Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. 651, 161–165 (2011). https://doi.org/10.1016/j.nima.2010.11.191
- [95] I. Harsányi, A. Horváth, Z. Kis et al., Assessment of neutron-induced



activation of irradiated samples in a research reactor. Nucl. Eng. Technol. 55, 1036-1044 (2023). https://doi.org/10.1016/j.net.2022.11.004

[96] Y.G. Yao, C.J. Xiao, Y.Q. Li, et al., Research on Neutron Activation Analysis Technology for Lunar Samples. Chinese Inorganic Analytical Chemistry 14, 93-103 (2024). doi:doi:10.3969/j.issn.2095-1035.2024.01.011 (in Chinese)

[97] A.D. Dolgov, Neutrinos in cosmology. Phys. Rep. 370, 333–535 (2002). https://doi.org/10.1016/S0370-1573(02)00139-4

[98] A.P. Serebrov, V.G. Ivochkin, R.M. Samoylov et al., Experiment for search for sterile neutrino at SM-3 reactor. Phys. Part. Nuclei 47, 1014–1023 (2016). https://doi.org/10.1134/s1063779616060228

[99] J. Ashenfelter, A.B. Balantekin, C. Baldenegro et al., The PROSPECT reactor antineutrino experiment. Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. (2019). https://doi.org/10.1016/j.nima.2018.12.079 [100] J. Ashenfelter, B. Balantekin, C.X. Baldenegro et al., Background radiation measurements at high power research reactors. Nucl. Instrum. Meth. Phys. Res. Sect. A Accel. Spectrometers Detect. Assoc. Equip. 806, 401-419 (2016). https://doi.org/10.1016/j.nima.2015.10.023

[101] N. Romanova, M. Aitkulov, K. Kisselyov et al., Neutron screen method for axial uniformity in silicon ingots for neutron transmutation doping at WWR-K reactor. Nucl. Energy (2023). https://doi.org/10.1016/j.anucene.2023.109697 [102] M. Komeda, T. Obara, Core designs based on research reactors for neutron transmutation doping of silicon. Ann. Nucl. Energy 65, 338–344 (2014). https://doi.org/10.1016/j.anucene.2013.11.035

[103] G.P. Gaidar, P.I. Baranskii, Thermoelectric properties of transmutation doped silicon crystals. Phys. B Condens. Matter 441, 80-88 (2014). https://doi.org/10.1016/j.physb.2014.02.011

[104] P. Coghi, T. Fazal, N.S. Hosmane et al., Diagnostic and theranostic technologies used in boron neutron capture therapy-A brief review. Inorg. Chem. Commun. 159, 111698 (2024). https://doi.org/10.1016/j.inoche.2023.111698 [105] M. Purohit, M. Kumar, Boron neutron capture therapy: History and recent advances. Mater. Today Proc. (2022). https://doi.org/10.1016/j.matpr.2022.12.181 [106] W. Xu, J. Li, H. Xie et al., Conceptual design and safety characteristics of a new multi-mission high flux research reactor. Nucl. Sci. Tech. 34, 34 (2023). https://doi.org/10.1007/s41365-023-01191-6

Note: Figure translations are in progress. See original paper for figures.

Source: ChinaXiv - Machine translation. Verify with original.